Author’s response to reviews

Title: Prune Belly Syndrome in Surviving Males can be caused by Hemizygous Missense Mutations in the X-linked Filamin A Gene

Authors:
Nida Iqbal (nida.iqbal3@gmail.com)
Thomas Jascur (tjascur@yahoo.com)
Steven Harrison (steven@broadinstitute.org)
Angelena Edwards (Angelena.edwards@utsouthwestern.edu)
Luke Smith (Luke.Smith@utsouthwestern.edu)
Erin Choi (Erinchoi123@utexas.edu)
Michelle Arevalo (Mkarvelo1@gmail.com)
Catherine Chen (Catherine.j.chen@utsouthwestern.edu)
Shaohua Zhang (shaohua.zhang@utsouthwestern.edu)
Adam Kern (adamjkern@gmail.com)
Angela Scheuerle (angela.scheuerle@utsouthwestern.edu)
Emma Sanchez (emma.sanchez@childrens.com)
Chao Xing (chao.xing@utsouthwestern.edu)
Linda Baker (linda.baker@childrens.com)

Version: 2 Date: 29 Jan 2020

Author’s response to reviews:

1. Abstract

Please ensure that the Abstract entered into your manuscript is the same as that entered into the submission system. - We have completed this
2. Corresponding author email - this has been corrected.

3. Availability of data and materials - the data has been uploaded to Clinvar and accession numbers added to the manuscript.

4. Role of funding bodies

Please provide more details regarding the specific role of each of the funding bodies in your study. If the funding bodies only provided the financial means to allow the authors to carry out the study, please state “The funding bodies played no role in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript” and include this statement in the Funding section of your manuscript. - this has been addressed.

5. Web Resources

Please remove the section entitled ‘Web Resources’ from your manuscript. - this has been removed

6. Figure 4

a) Please ensure that you include figure legends for each of the sub-figures (A, B, C, D) of Figure 4 in your manuscript.

b) As per our submission guidelines, we require that all figures, tables and additional files be cited in the text in sequence in a manuscript. Figure 4A is currently not cited in your manuscript. Please ensure that it is appropriately cited within the main body of text in your manuscript.
This has been addressed.

7. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Should you wish to respond to these revision requests, please put your responses to the reviewers'/editors’ comments in the Response to Reviewers box in Editorial Manager. Please do not upload a separate letter.

This has been addressed.